Author's response to reviews

Title: Economic evaluation of Community Level Interventions for Pre-Eclampsia (CLIP) in South Asian and African countries: a study protocol

Authors:
Asif R Khowaja Mr (araza@cw.bc.ca)
Craig Mitton Dr (craig.mitton@ubc.ca)
Stirling Bryan Dr (stirling.bryan@ubc.ca)
Peter von Dadelszen Dr (pdadelszen@cw.bc.ca)

Version: 4 Date: 12 May 2015

Author's response to reviews:

Dear Editor,

We have incorporated in our manuscript, the good reporting practices as stipulated by the Consolidated Health Economic Evaluation Reporting Standards (CHEERS). The CHEERS checklist has been uploaded online along with the revised manuscript.

We hope that the revised documents would be helpful for the editorial review.

Thank you

Craig Mitton